Author's response to reviews

Title: Improved Sensitivity of an Interferon-Gamma Release Assay (T-SPOT.TB) in Combination with Tuberculin Skin Test for the Diagnosis of Latent Tuberculosis in the Presence of HIV Infection

Authors:

Luigia Elzi (elzil@uhbs.ch)
Ingrid Steffen (ingrid.steffen@unibas.ch)
Hansjakob Furrer (hansjakob.furrer@insel.ch)
Jan Fehr (jan.fehr@usz.ch)
Matthias Cavassini (matthias.cavassini@chuv.ch)
Bernard Hirschel (bernard.hirschel@hcuge.ch)
Matthias Hoffmann (matthias.hoffmann@kssg.ch)
Enos Bernasconi (enos.bernasconi@eoc.ch)
Stefano Bassetti (stefano.bassetti@spital.so.ch)
Manuel Battegay (mbattegay@uhbs.ch)

Version: 2 Date: 29 June 2011

Author's response to reviews:

Dear Editor,

We are pleased to submit a revision of our manuscript entitled "Improved Sensitivity of an Interferon-Gamma Release Assay (T-SPOT.TB) in Combination with Tuberculin Skin Test for the Diagnosis of Latent Tuberculosis in the Presence of HIV Infection".

As suggested, we added a section on "Ethical approval" in the Methods paragraph. All changes are marked in yellow.

Thank for kindly considering our manuscript.

Yours sincerely,

Manuel Battegay